Title: Typhoid fever as a cause of opportunistic infection: case report

Authors:

Claudia Colomba (claudia.colomba@libero.it)
Laura Saporito (laura.sapo@tin.it)
Laura Infurnari (infurnari.laura@libero.it)
Salvatore Tumminia (claudia.colomba@libero.it)
Lucina Titone (titonel@unipa.it)

Version: 5 Date: 17 December 2005

Author’s response to reviews: see over
BMC Infectious Diseases,
BioMed Central Ltd,
Middlesex House, 34-42 Cleveland Street,
London W1T 4LB, UK.

Dear Sir,

We are delighted to know that the editors of BMC Infectious Diseases have approved the publication of our manuscript titled “Typhoid fever as a cause of opportunistic infection: case report”.

All the authors are responsible for reported research. They have participated in the concept and design; analysis and interpretation of data; drafting or revising of the manuscript.

There are not any affiliation, financial agreement, or other involvement of any author with the companies whose products figures prominently in the submitted typescript.

The authors ensure that no material submitted infringes existing copyrights, or the rights of a third party.

The material presented has not been published previously, and will not be submitted elsewhere for publication as long as it is under consideration of BMC Infectious Diseases.

We send you the final version of the revised manuscript. We did not do any changes to the text.

Yours faithfully,

Claudia Colomba

Corresponding author

Claudia Colomba, Istituto di Patologia Infettiva e Virologia - Università di Palermo – Piazza Montalto n. 8, 90134 Palermo, Italy (claudia.colomba@libero.it)